Title: A boy presenting with idiopathic eosinophilic parotitis: a case report.

Authors:

Franco Frati (frati.f@stallergenes.it)
Rachele Boccardo (lele.raissa@gmail.com)
Silvia Scurati (scurati.s@stallergenes.it)
Matteo Gelardi (gelardim@inwind.it)
Cristoforo Incorvaia (cristoforo.incorvaia@gmail.com)

Version: 3 Date: 31 May 2011

Author's response to reviews: see over
Reviewer 1

The case is very interesting and well reported. It would take advantage from a linguistic revision from an english medical editor.

*The manuscript was professionally edited.*

Reviewer 2

It's a very interesting case report.

If it's possible I should make mention about immunological examination as FAN/ANA or anti-DNA for a possible link with an autoimmune disease.

*We added that anti-nuclear antibodies and anti-DNA antibodies were included in the blood examination, with negative results.*